Author's response to reviews

Title: The intron 4c allele of the NOS3 gene is associated with ischemic stroke in African Americans

Authors:

   Raji P Grewal (rgrewal@solarishs.org)
   Ana VC Dutra (dutracacd@hotmail.com)
   Yichu C Liao (yichu.liao@gmail.com)
   Hank Juo (hjuo@kmu.edu.tw)
   Nikolaos IH Papamitsakis (papamitsakis@yahoo.com)

Version: 4 Date: 6 December 2007

Author's response to reviews: see over
Biomed Central Editorial Production Team:

I have made the following revisions in this version of the manuscript:

1-made the changes to the author list including the removal of qualifications and adding the country of affiliation.

2-placed each section in the abstract in paragraph form.

3-reformatted the references to comply with journal style.

4-eliminated page 15.

I hope these changes are acceptable.

Yours sincerely:

Raji P. Grewal MD